Author’s response to reviews

Title: FUMEPOC Early detection of chronic obstructive pulmonary disease in smokers

Authors:

Vicente Gil-Guillén (atencion.primaria@umh.es)
Domingo Orozco-Beltrán (dorozcobeltran@gmail.com)
Concepcion V Carratala Munuera (atencion.primaria@umh.es)
Carlos Plaza-Sirvent (biokalos@hotmail.com)
Patricia Lorca-Amorrich (patri_lorca@hotmail.com)
Adriana López-Pineda (adriannalp@hotmail.com)
María P Vela-Troncoso (aki_pilar2002@hotmail.com)
Juan J Soler (jjsoler@telefonica.net)
Manuel Yarza-Cañellas (yarza_man@gva.es)
Antonio Fernández (tonifdz@yahoo.es)
Luis Rosado-Bretón (rosado_lui@gva.es)
Carmen Olivares-Bautista (olivares.c@hotmail.es)
Alejandro Muñoz-Fernández (alejandro_fm@ono.com)
FUMEPOC Study Research Team [FSRT] (atencion.primaria@umh.es)

Version: 6 Date: 19 May 2011

Author’s response to reviews:

Dear editorial board of BMC Public Health,

We have submitted our revised manuscript with the requested changes.

We have made changes in the author list and now the names are written exactly as they are in the submission system, both in style and order. Moreover we have added the name of group within the list of authors and in the submission system.

Also we have removed the names of the members of the group from the title page and we have placed them in the Acknowledgements section.

Regarding the Abstract, we have changed one of the headings in the submission system and now the abstract is identical in the manuscript and the submission system.

In addition, you will see that authors´ contributions is formatted into a paragraph, as you required.

Finally, we have uploaded the figure file again and it is not included the title or the figure number.

Do not hesitate to contact us if you have any suggestion.

Kind regards,
Vicente Gil-Guillen on behalf of the authors